# Yann Renard\* Fabien Lotte

INRIA Rennes 35042 Rennes Cedex France

**Guillaume Gibert** INSERM U821 Lyon, France

# Marco Congedo

INPG Gipsa-Lab Grenoble, France

Emmanuel Maby

INSERM U821 Lyon, France

Vincent Delannoy INRIA Rennes

France

**Olivier Bertrand** 

INSERM U821 Lyon, France

# Anatole Lécuyer\*

INRIA Rennes 35042 Rennes Cedex France

# OpenViBE: An Open-Source Software Platform to Design, Test, and Use Brain-Computer Interfaces in Real and Virtual Environments

# Abstract

This paper describes the OpenViBE software platform which enables researchers to design, test, and use brain-computer interfaces (BCIs). BCIs are communication systems that enable users to send commands to computers solely by means of brain activity. BCIs are gaining interest among the virtual reality (VR) community since they have appeared as promising interaction devices for virtual environments (VEs). The key features of the platform are (1) high modularity, (2) embedded tools for visualization and feedback based on VR and 3D displays, (3) BCI design made available to non-programmers thanks to visual programming, and (4) various tools offered to the different types of users. The platform features are illustrated in this paper with two entertaining VR applications based on a BCI. In the first one, users can move a virtual ball by imagining hand movements, while in the second one, they can control a virtual spaceship using real or imagined foot movements. Online experiments with these applications together with the evaluation of the platform computational performances showed its suitability for the design of VR applications controlled with a BCI. OpenViBE is a free software distributed under an open-source license.

# I Introduction

One of the keys to a great immersion feeling with virtual reality (VR) is the ease of interaction with virtual environments (VE). Recently, a new method has emerged: interacting through cerebral activity, using a brain–computer interface (BCI; Leeb et al., 2007, 2006; Lécuyer et al., 2008). Such an interface is a communication system that enables a user to send commands to a computer by means of variations of brain activity, which is in turn measured and processed by the system (Wolpaw, Birbaumer, McFarland, Pfurtscheller, & Vaughan, 2002). BCI are currently following the path laid down by haptic devices a few years ago (Burdea, 1996) by providing a completely new way of conceiving interaction with computers and electronic devices through the "interaction by thought" concept. The BCI technology is rapidly improving,

Presence, Vol. 19, No. 1, February 2010, 35–53 © 2010 by the Massachusetts Institute of Technology

<sup>\*</sup>Correspondence to yann.renard@irisa.fr and anatole.lecuyer@irisa.fr.

and several interesting applications using BCI have already been developed for navigating or interacting with virtual environments (Leeb et al., 2007; Lécuyer et al.; Friedman et al., 2007), or for video games (Krepki, Blankertz, Curio, & Müller, 2007; Nijholt, 2009).

However, designing BCI-based interaction devices requires expertise in a broad range of domains, ranging from neurophysiology, signal processing, and interaction, to computer graphics or computer programming which represents a challenging multidisciplinary task. A general purpose software platform that provides the necessary functionalities to easily design BCI and connect them with VR would foster research in the domain and democratize the use of BCI in real and virtual environments.

In this paper, we present the OpenViBE platform, a novel, free, and open source platform to design and tune BCI systems and connect them with real and virtual environments. This paper is organized as follows. Section 2 covers a short state of the art of existing BCI platforms, and Section 3 describes the features of our platform. Section 4 presents the range of users our system targets. Sections 5 and 6 detail respectively the design of a BCI with OpenViBE and the tools we provide. Section 7 is dedicated to the connection with VR, and Section 8 details the platform internals. Finally, some examples of BCI implementations, performance, and the current state of the platform are presented respectively in Sections 9, 10, and 11. The paper ends with a general conclusion.

#### 2 Related Work: Existing BCI Software

Several software programs for off-line and online analysis of EEG and biomedical signals are available. They are briefly reviewed in Schlögl, Brunner, Scherer, and Glatz (2007). However, these programs do not include all the necessary functionalities for designing a BCI.

In the freeware community, only three programs enclose the necessary functionalities for real-time BCI designs: BioSig (Schlögl et al., 2007; thanks to the "rts-BCI" package), BCI2000 (Mellinger & Schalk, 2007), and BCI++ (Maggi, Parini, Perego, & Andreoni, 2008).

BioSig is an open-source software library for biomedical signal processing and more specifically for BCI research (Schlögl et al., 2007). It is a toolbox for Octave and MATLAB that offers several data management modules, data import and export, artifact processing, quality control, feature extraction algorithms, classification methods, and so on. It also offers rapid prototyping of online and real-time BCI with the rtsBCI package using MATLAB/Simulink.

BCI2000 is a general purpose system for BCI research (Mellinger & Schalk, 2007). This software is not open source, but its source code and executables are available for free for nonprofit research and educational purposes. BCI2000 is a C++ software program that proposes to build an online and real-time BCI by assembling four modules: the source module, for data acquisition and storage; the signal processing module, that encompasses the preprocessing, feature extraction, and classification of brain activity; the user application module, with which the user interacts; and finally, the operator interface for data visualization and system configuration. Interestingly, BCI2000 also provides tools for off-line analysis of data within the MARIO software. Recently, another BCI software platform has been proposed: BCI++ (Maggi et al., 2008). This software is a C/C++ framework for designing BCI systems and experiments. BCI++ also includes some 2D/3D features for BCI feedback. However, it should be noted that this platform is not completely open source.

A comparison of these software programs with Open-ViBE is provided in Section 3.1.

#### **3 OpenViBE Features**

OpenViBE is a free and open source software platform for the design, test and use of BCIs. The platform consists of a set of software modules that can be easily and efficiently integrated to design BCI for both real and VR applications. Key features of the platform are: modularity and reusability, applicability for different types of users, portability, and connection with VR.

- Modularity and Reusability Our platform is a set of software modules devoted to the acquisition, preprocessing, processing, and visualization of cerebral data, as well as to interaction with VR displays. OpenViBE, being general purpose software, implies that users are able to easily add new software modules in order to fit their needs. This is ensured thanks to the box concept, an elementary component in charge of a fraction of the whole processing pipeline, that allows users to develop reusable components, reduces development time, and helps to quickly extend functionalities.
- **Different Types of Users** OpenViBE is designed for different types of users: VR developers, clinicians, BCI researchers, and so on. Their various needs are addressed and different tools are proposed for each of them, depending on their programming skills and their knowledge of brain processes.
- **Portability** The platform operates independently from the different software targets and hardware devices. It includes an abstract level of representation, allowing it to run with various input devices, such as EEG or MEG. It can run on Windows and Linux operating systems and also includes different data visualization techniques. Finally, it is based on free and portable software (e.g., GTK+,<sup>1</sup> IT++,<sup>2</sup> GSL,<sup>3</sup> VRPN,<sup>4</sup> GCC<sup>5</sup>).
- **Connection with VR** Our software can be integrated with high-end VR applications. OpenViBE acts as an external peripheral to any kind of real and virtual en-

1. The Gnome ToolKit is a highly usable, feature rich toolkit for creating graphical user interfaces which boasts cross platform compatibility and offers an easy to use API. More information can be found at http://www.gtk.org.

2. IT++ is a C++ library of mathematical, signal processing, and communication routines. More information can be found at http://sourceforge.net/apps/wordpress/itpp .

3. The GNU Scientific Library is a numerical library for C and C++ programmers. More information can be found at http://www.gnu.org/software/gsl.

4. The Virtual-Reality Peripheral Network is a set of classes within a library designed to implement an interface between application programs and the set of physical devices used in a virtual reality system. More information can be found at http://www.cs.unc.edu/Research/ vrpn .

5. The GNU Compiler Collection is a compiler that supports a wide range of architectures. More information can be found at http://gcc.gnu.org .

vironment. It also takes advantage of VR displays thanks to a light abstraction of a scenegraph management library, allowing users to visualize cerebral activity in an understandable way or to provide incentive training environments (e.g., for neurofeedback).

### 3.1 Comparison with Other BCI Platforms

In comparison to other BCI software, the Open-ViBE platform is highly modular. It addresses the needs of different types of users (should they be programmers or nonprogrammers) and proposes a user-friendly graphical language that allows nonprogrammers to design a BCI without writing a single line of code. In contrast, all other BCI platforms require some degree of programming skills to design a new real-time BCI from scratch. Furthermore, their modularity is coarser (except for BioSig), hence restricting the range of possible designs.

OpenViBE is also portable, independent of the hardware or software, and is entirely based on free and open source software. In comparison, among other real-time BCI platforms, only BioSig is fully open source but the rtsBCI package needed for online and real-time BCI requires MATLAB/Simulink, which is non-free and proprietary software.

OpenViBE proposes to generate online scenarios (step 3 in Figure 1) automatically from off-line analysis (step 2 in Figure 1). Finally, in contrast to other platforms, OpenViBE is well suited for VR applications as it provides several embedded tools to design innovative VR displays and feedback as well as to perform 3D visualization of brain activity in real time. Furthermore, Open-ViBE can also be used as a device for any VR application.

#### 4 Different Types of Users

OpenViBE has been designed for four types of users. The first two types, the developer and the application developer, are both programmers; the last two types, the author and the operator, do not need any programming skills.



Figure 1. Designing a BCI with OpenViBE.

- The Developer (Programmer): The developer has the possibility to add new functionalities and test original pieces of software in OpenViBE. To that end, OpenViBE is delivered with a complete software development kit (SDK). This SDK provides access to functionalities at different levels depending on the task at hand. There are two main categories of developers: first, the kernel developers who enhance and modify kernel functionalities (see Section 8.2); second, plug-in developers who create new additional modules (see Section 8.3).
- **The Application Developer (Programmer):** The application developer uses the SDK to create standalone applications, using OpenViBE as a library. Such applications range from new tools such as the visual scenario editor described in Section 6, to external VR applications with which the BCI user can interact. Such VR applications are presented in Section 7.
- The Author (Non-programmer): The author uses the visual scenario editor (see Figure 2) to arrange existing boxes to form a scenario. The author configures these boxes and the scenario in order to produce a complete, ready-to-use BCI system. The author is aware of the internals of our platform as well as of BCI systems and is familiar with basic signal processing. The author is also aware of the



**Figure 2.** The OpenViBE designer with a sample scenario. The tool enables the graphical design of a BCI system by adding and connecting boxes representing processing modules without writing a single line of code.

interaction paradigm to use. However, he or she does not need strong computer programming skills because there are dedicated tools to perform the tasks (see Section 6).

The Operator (Non-programmer): The operator

generally would be a clinician or a practitioner, and is neither a computer expert nor an OpenViBE expert. The operator is in charge of using and running the prebuilt scenarios of the author. The operator then simply runs the scenario. The operator is aware of how the BCI system should and can work, and monitors the execution of the BCI system thanks to dedicated visualization components. He or she has an understanding of neurophysiological signals and can help the BCI user to improve control over the BCI system.

Finally, another role should be considered: the BCI user. The BCI user generally wears the brain activity acquisition hardware (e.g., an EEG cap) and interacts with an application by means of mental activity. The application could be, for instance, a neurofeedback training program, a video game in virtual reality, a remote operation in augmented reality, and so on. While the user does not directly use the OpenViBE platform, he or she implicitly takes advantage of its features.

#### 5 How to Design a BCI with OpenViBE

Designing and operating an online BCI with our software follows a rather universal way of doing so (Wolpaw et al., 2002). Three distinct steps are required (see Figure 1). In the first step, a training data set must be recorded for a given subject who performs specific mental tasks. The second step consists of an off-line analysis of these records with the goal of finding the best calibration parameters (e.g., optimal features, relevant channels, etc.) for this subject. The last step consists of using the BCI online in a closed loop process. Optionally, iterations can be done on data acquisition and off-line training in order to refine the parameters.

The online loop (third step) is common to any BCI and it is composed of six phases: brain activity measurements, preprocessing, feature extraction, classification, translation into a command, and feedback (see Figure 1).

- **Brain Activity Measurements:** This step consists of measuring the brain activity of the BCI user. To date, about half a dozen different kinds of brain signals have been identified as suitable for a BCI, that is, they are easily observable and controllable (Wolpaw et al., 2002). Measuring the brain activity for a BCI system is mainly performed using electro-encephalography (EEG) since it is a cost-effective and noninvasive method which provides high temporal resolution (Wolpaw et al., 2002). Our software already supports various EEG acquisition devices but also magnetoencephalography (MEG) machines (see Section 11 for a list of supported devices).
- **Preprocessing:** The preprocessing step aims at removing the noise from the acquired signals and/or at enhancing a specific brain signal (Bashashati, Fatourechi, Ward, & Birch, 2007). For example, our software proposes different kinds of preprocessing algorithms such as temporal filters and spatial filters (independent component analysis, surface Laplacian, etc.).
- Feature Extraction: Once signals have been preprocessed, features can be extracted. These features consist of a few values that describe the relevant information embedded in the signals (Bashashati et al., 2007) such as the power of the signals in specific frequency bands (Pfurtscheller & Neuper, 2001). These features are then gathered into a vector called the feature vector. Examples of features available in OpenViBE include band power features or power spectral densities.
- **Classification:** The feature vector is fed into an algorithm known as the classifier. A classifier assigns a class to each feature vector, this class being an identifier of the brain signal that has been recognized. In general, the classifier is trained beforehand using a set of feature vectors from each class. An example of classifier used for BCI would be linear discriminant analysis (Lotte, Congedo, Lécuyer, Lamarche, & Arnaldi, 2007). It should be noted that, due to the high variability and noisiness of EEG signals, classification rates of 100% are very rarely attained, even for a BCI using only two men-

tal states. OpenViBE proposes several classifiers such as linear discriminant analysis (Lotte, Congedo, et al.) or fuzzy inference systems (Lotte, Lécuyer, Lamarche, & Arnaldi, 2007).

- **Translation into a Command:** Once the class of the signal has been identified, it can be associated to a command that is sent to a computer in order to control, for instance, a robot (Millán, 2008) or a prosthesis (Wolpaw et al., 2002). The number of possible commands in current EEG-based BCI systems typically varies between one and four.
- Feedback: Finally, feedback should be provided to the user so that he or she can determine whether the brain signal was performed correctly. This is an important step as it helps the user to control his or her brain activity (Lotte, Renard, & Lécuyer, 2008; Neuper, Scherer, Wriessnegger, & Pfurtscheller, 2009). Feedback can be simple visual or audio cues, for example, gauges. To this aim, our software proposes classical raw signal, spectra, time/frequency visualization modules. Alternatively, more advanced feedback can be provided such as the modification of a virtual environment (Leeb et al., 2007) to which OpenViBE sends commands.

#### 6 Tools

Our system includes a number of useful tools for its various users: the acquisition server, the designer, 2D visualization tools, and sample scenarios of BCI or neurofeedback.

The acquisition server provides a generic interface to various kinds of acquisition machines, for example, EEG or MEG systems. Such an abstraction allows the author to create hardware independent scenarios, thanks to the use of a generic acquisition box. This box receives the data via the network from the acquisition server, which is actually *connected* to the hardware and transforms these data in a generic way. The way the acquisition server gets *connected* to the device mostly depends on the hardware manufacturer's way to access the device. Some devices will be shipped with a specific SDK, while others will propose a communication protocol over a network/serial/USB connection. Finally, some devices will need proprietary acquisition software that delivers the measures to the acquisition server.

The designer is mainly dedicated to the author and enables the author to build complete scenarios based on existing software modules using a dedicated graphical language and a simple graphical user interface (GUI) as shown in Figure 2. The author has access to a list of existing modules in a panel, and can drag and drop them in the scenario window. Each module appears as a rectangular box with inputs (on top) and outputs (at the bottom). Double-clicking on a box displays its configuration panel. Boxes are manually connectable through their inputs and outputs. The designer also allows the author to configure the arrangement of visualization windows (i.e., visualization modules included in the scenario). An embedded player engine allows the author to test and debug a scenario in real time. In doing so, the author can receive continuous feedback on box status and processing times. Such feedback may be useful to balance the computational load.

The **2D** visualization features of the platform are available as specific boxes and include brain activity related visualizations. These boxes can access all the platform functionalities, and particularly the whole stream content for the connected inputs. Most 2D visualization boxes display input data in a widget and do not produce output. Our system offers a wide range of visualization paradigms such as raw signal display, gauges, power spectrum, time-frequency map, and 2D topography in which EEG activity is projected on the scalp surface in two dimensions (see Figure 3). OpenViBE also provides a visualization tool that displays instructions to a user according to the protocol of the famous Graz motor imagerybased BCI (Pfurtscheller & Neuper, 2001).

Existing and preconfigured ready-to-use scenarios are proposed to assist the author. As the creation of new scenarios has gotten faster and easier, the number of available scenarios is expected to rapidly increase. Currently, five complete scenarios are available.



**Figure 3.** Examples of 2D displays: raw signals and time-frequency map display widgets.

- Hand Motor Imagery-Based BCI: This scenario allows the use of OpenViBE as an interaction peripheral using imagined movements of the left and right hand. This scenario is inspired from the wellknown Graz-BCI of the Graz University (Pfurtscheller & Neuper, 2001; see Section 9).
- Self-Paced BCI Based on Foot Movements: This scenario represents a BCI based on real or imagined foot movements that can be used in a self-paced way. This means the subject can interact with the application at any time, contrary to most existing BCIs (see Section 9).
- Neurofeedback: This scenario shows the power of the brain activity in a specific frequency band, and helps a subject in the task of self-training to control that power.
- Real-Time Visualization of Brain Activity in 2D/3D: This scenario enables the user to visualize his or her own brain activity evolving in realtime on a 2D or 3D head model. This scenario can be used together with inverse solution methods (Baillet, Mosher, & Leahy, 2001), in order to visualize the brain activity in the whole brain volume, not only on the scalp surface, as in Arrouët et al. (2005; see Figure 6 to be discussed later).
- **P300 Speller:** This scenario implements the famous P300 speller (Farwell & Donchin, 1988; Donchin, Spencer, & Wijesinghe, 2000), an application that enables a user to spell letters by using only brain activity, and more precisely the event

related potential known as the P300 (Wolpaw et al., 2002; see Figure 4 and Figure 6 later in this paper). It should be noted that OpenViBE can also be used to design other P300-based BCI, and not only the P300 speller. Interested readers may refer to Sauvan, Lécuyer, Lotte, and Casiez (2009) for another example of a P300-based BCI designed with OpenViBE.

Figure 5 summarizes how users interact with software and hardware components. For example, the operator uses both the acquisition server and the designer; the brain activity acquisition device feeds the acquisition server and the analysis is performed on the computer hosting the designer. This results in a dedicated embedded visualization that monitors the user's brain activity.

#### 7 Connection with Virtual Reality

The platform includes a number of embedded 3D visualization widgets and is able to interact with external VR applications thanks to standard communication protocols. However, OpenViBE does not aim at offering a complete set of scenegraph managing capabilities, nor at embedding a VR application builder. Consequently, OpenViBE users should be aware of what the platform is in charge of, and what is left to external applications communicating with OpenViBE.

# 7.1 OpenViBE as an Interaction Device for External Virtual Reality Applications

Our platform can be used as an interaction peripheral for any general purpose application in real and virtual environments. As such, some of the data processed by the scenario need to be exposed to the outside world. There are two ways this goal can be achieved.

One way to meet this goal is to propose specific boxes that can expose parameters in a "considered as standard" way. For example, the platform includes a VRPN module (Taylor et al., 2001) that acts as a server and sends analogic and button values. This is a convenient way to interact with existing VR applications. The ad-



Figure 4. Examples of visualization widgets available in OpenViBE. Left: The P300 speller. Right: 2D visualization of brain activity in real time, on the scalp.



Figure 5. Relations between users, hardware, and software components.

vantage stands in that VR application developers do not have to perform major modifications on their application to have it controlled by a BCI user. Examples of VR applications using OpenViBE and the VRPN plug-in are given in Section 9.

The other way to meet the goal is to build an application using the platform as a third-party peripheral management library. The developer has access to the whole exposed data and is able to process and display it in his or her own application.

# 7.2 OpenViBE for Direct Visualization and Interaction with 3D Models

In order to perform 3D visualization, with or without VR displays, the OpenViBE kernel hides a scenegraph manager and exposes a number of functionalities such as color, position, transparency and scale settings for 3D objects, as well as mesh management capabilities. This allows developers to easily and quickly develop 3D plug-ins using a simplified 3D



**Figure 6.** 3D display of brain activity. Left: 3D topographic display of brain activity in real time, on the scalp. Right: Voxel reconstruction of brain activity inside the brain, based on scalp measures.

Application Programming Interface (API). This API offers the required functionalities to load and dynamically modify a 3D scene based on the input data and allows direct visualization and interaction with 3D models.

# 7.3 OpenViBE for Real-Time Visualization of Brain Activity

OpenViBE is also used to visualize brain activity in real time or to get immersive neurofeedback. In order to achieve this, the scenegraph manager is used by several visualization widgets. Figure 6 shows two examples of what our platform offers in terms of embedded 3D widgets for real-time visualization of brain activity: a 3D topographic map which displays the recorded potentials mapped onto a 3D head, and a voxelized reconstruction of the inside brain activity, based on scalp measures.

#### 8 **OpenViBE Internals**

This section describes the software architecture of the platform. In order to design an extensible software, we followed the approach of already existing and widely used VR software such as Virtools (http://www.virtools. com). In this software package, the classical kernel and plug-in architecture ensures maximum extensibility. A new plug-in can be dynamically added and used by the kernel for the applications' benefit, without the need to rebuild the application or the kernel itself. Additionally, composing scenarios based on elementary components ensures maximum flexibility and reusability.



Figure 7. Software architecture.

Therefore, each application of our platform relies on a common kernel that delegates tasks to a number of dedicated plug-ins as shown in Figure 7. Moreover, the kernel offers the concept of the box, allowing for the creation of powerful tools such as the designer authoring tool. Each of these components is presented in the following sections.

#### 8.1 The Box Concept

The box is a key component of the platform. It consists of an elementary component in charge of a fraction of the whole processing pipeline. It exposes inputs and outputs to other boxes. Each box can be notified on clock ticks and upon input data arrival. The behavior of a box can be adapted to the needs of each algorithm (e.g., acquisition algorithms typically react to clock signals whereas processing algorithms typically react to input arrival). The characteristics and constraints that are common to all boxes include reasonable granularity to allow quick software components rearrangement. Newly developed boxes are immediately available to the user thanks to the plug-in system (see Section 8.3).

#### 8.2 The Kernel

The kernel provides global services to applications through several managers, each of them providing a set of specialized services. For example, the plug-in manager makes the platform extensible. This manager is able to dynamically load plug-in modules (e.g., .dll files under Windows, or .so files under Linux) and collect extensions from them, such as scenario serializers, algorithms, and boxes (see Section 8.3). The plug-in system allows for the quick and efficient expansion of functionalities. The communication interface between these extensions and the kernel itself is defined so that they can easily be shared, used, and replaced when needed.

Another example is the scenario manager, which helps creating and configuring scenarios. For instance, the manager can add or remove boxes, change their settings, and connect them together. The scenario manager can handle multiple scenarios simultaneously. The designer authoring tool takes advantage of this in order to edit them in multiple tabs.

Finally, the visualization manager is responsible for displaying 2D or 3D graphical information and setting the position and size of the displays in a window. Indeed, multiple visualization windows may be used. The windows arrangement in space is done by the visualization manager at editing time, thanks to the designer application, and saved to a file. Basic signal display windows are provided with a 2D rendering context (see Figure 3), while more advanced rendering is performed thanks to the encapsulated 3D library (see Section 7).

Several other managers exist, such as the player manager for an easy setup of a runtime session, the configuration manager for a convenient way to configure the whole platform with text files, and the type manager which ensures coherency and possibly conversions of all data types (e.g., box settings or connectors). Interested readers will find more information about those managers in the software documentation (INRIA, 2010).

#### 8.3 Plug-ins

Our platform includes three different families of plug-in: driver plug-ins, algorithm plug-ins, and box plug-ins.

The driver plug-ins allow the user to add acquisition devices to the acquisition server. A driver basically reads

the signal from the device through a specific SDK or a physical connection, and injects this signal into Open-ViBE in a generic way. The rest of the processing pipeline is therefore independent of the acquisition hardware.

The algorithm plug-ins are a generic abstraction for any extension that could be added to the platform (e.g., add new feature extraction or signal processing methods). Algorithms are the developer's atomic objects. The developer may compose several algorithms in order to achieve a complex task. This kind of plug-in allows the user to massively share and reuse software components, even in an off-line context where time is handled at a different scale (e.g., EEG file reading or signal visualization widgets).

The box plug-ins are the software components each box relies on. Boxes are the author's atomic objects. The developer describes them in a simple structure that notably contains the box prototype (its name, input/ output connectors and settings). The box is responsible for the actual processing, that is, it reads from inputs, computes data to produce a result, and writes to outputs. The box generally combines several algorithm plug-ins together to perform its processing. This ensures fast development thanks to the reusability of components.

Additionally, it should be stressed that a specific box is available to developers: a box that accepts MATLAB code. This box aims at providing a tool to quickly develop and test some algorithms using the MATLAB language. As soon as the prototype is functional, it can be implemented in C++ for better performance.

#### **9** Examples of Implementation

In this section, we illustrate the capabilities of our software and the way to use it as an interaction device with two immersive applications: the "handball" and the "use-the-force" applications. The description of the handball application includes a special emphasis on the way OpenViBE is used to design the BCI and its associated scenarios.

#### 9.1 The Handball Application

The handball VR application is an immersive 3D game in which the user can control a virtual ball by using a BCI based on imagined hand movements. The objective of the game is to move the ball into a goal cage. As such, this application enables the researcher to illustrate the use of OpenViBE for the design of a very popular kind of BCI, namely a motor imagery-based BCI (Pfurtscheller & Neuper, 2001), and its use for interaction with a VR application. This section briefly describes the BCI system and the VR game, then it details how OpenViBE is used in the implementation of this application and reports on an online experiment with real subjects.

9.1.1 BCI System. For this application, we used a motor imagery-based BCI which is inspired by the well-known Graz BCI (Pfurtscheller & Neuper, 2001). Such BCI have been used successfully with several VR applications (Friedman et al., 2007; Leeb et al., 2006). With this system, the user has to perform imagined movements of the left or right hand to generate the brain signals expected by the BCI. It is established that performing an imagined hand movement triggers EEG power variations in the  $\mu$  ( $\approx$ 8–13 Hz) and  $\beta$  ( $\approx$ 13–30 Hz) frequency bands, over the motor cortices (Pfurtscheller & Neuper, 2001). Consequently, to identify these specific variations, the BCI uses logarithmic band power (BP) for feature extraction. Such features are simply computed by band-pass filtering the signal in subject-specific frequency bands (roughly in the  $\mu$  and  $\beta$ bands), squaring it, averaging it over a given timewindow, and computing its logarithm. Such features are extracted from the EEG channels located over the motor cortex. The generated feature vector is then passed to an efficient and widely used classifier, the linear discriminant analysis (LDA), which will identify the signal class, that is, either left or right, depending on the hand chosen for the imagined movement. In order to improve the performance of the BCI, we also used temporal and spatial filtering as preprocessing (see Section 9.1.3 for details).



**Figure 8.** The handball VR application. The user can move the virtual ball toward the left or right by imagining left or right hand movements.

9.1.2 Virtual Reality Game. The VE for this application was a virtual gymnasium equipped with a handball playing court and its two goal cages. The two goals were located on each side of the screen (see Figure 8); a virtual ball was also located in this VE and could be controlled by the user as part of the game. Each time the BCI system recognized an imagined left hand movement in the brain activity, an event was sent to the VE, and the ball rolled toward the left goal. By symmetry, the detection of an imagined right hand movement triggered the ball to roll toward the right goal. It should be noted that this application operated in a synchronous mode, which means the user could move the ball only during specific time periods, instructed by the system. As part of this game, the player's objective was to bring the ball into one of these two goals, as instructed by the application. More precisely, a game session was composed of 40 trials, among which 20 instructed the user to score in the left goal and 20 in the right goal. The order of the trials was randomized within a session. A trial was arranged as follows: at the beginning of the trial (t = 0 s), the ball was located at the center of the playing ground, that is, at the center of the screen. At t = 2 s, the ball color changed from red to green to indicate that the user

should get ready to perform motor imagery to move the ball. At t = 3.25 s, a downward pointing arrow appeared above one of the two goals to indicate the target goal. From t = 4 s to t = 8 s, the user can move the ball continuously by using motor imagery and should try to reach the target goal. At the end of the trial, the ball automatically goes back to the screen center. The user scores a point if, at the end of the trial, the ball is closer to the target goal than to the other. A trial is followed by a short rest period of random duration. It should be noted that the experimental paradigm used in this application is equivalent to that of the Graz BCI protocol (Pfurtscheller & Neuper, 2001).

**9.1.3 Implementation with OpenViBE.** As mentioned previously, before using a BCI, an off-line training phase is required in order to calibrate the system. This training phase needs a set of sample EEG signals. Consequently, the implementation of the BCI of this application is divided into four OpenViBE scenarios: three scenarios for the calibration of the BCI (acquisition of training data, selection of subject-specific frequency bands, and classifier training) and one scenario for the online use of the BCI.

Step 1: Acquisition of Training Data This phase aims at collecting training EEG data recorded while the subject performs imagined left or right hand movements. The scenario corresponding to this phase simply consists of the assembly of four boxes. The first one is a generic network acquisition box which acquires the recorded EEG signals. The second is a file writer box, which writes these EEG signals into a file using the general data format (GDF) (Schlögl, 2006). The next box is a visualization box which displays the instructions that the user will have to follow. These instructions can be to perform an imagined movement of the left or right hand, rest, or other mental actions. Finally, a stimulation box is used. This box generates events according to an XML file passed as a parameter. These events are sent to the visualization box, which will display the corresponding instructions, and to the file writer box, which will store

the events in order to know when the subject was asked to perform imagined movements. These events are generated according to the Graz BCI protocol (Pfurtscheller & Neuper, 2001).

Step 2: Off-line Training This phase consists of determining the optimal BCI parameters for the subject, that is, the optimal frequency bands for discriminating the two brain states using BP features, and the parameters of the LDA classifier. The optimal frequency bands are obtained using a statistical analysis on the training EEG signals, as in Zhong, Lotte, Girolami, and Lécuyer (2008) and Lotte, Lécuyer, et al. (2007). The LDA classifier is then trained on the BP features extracted from the EEG training data. This off-line training phase is decomposed into two scenarios: one for selecting the optimal frequency bands, and one for training the classifier. For these two scenarios, three specific boxes are necessary: a GDF file reader, in charge of reading the data recorded during the previous phase; a statistical analysis box that will estimate the best frequency bands in which to extract the BP features; and an LDA training box to train the classifier on these features. All obtained parameters are saved for further use, that is, during the online phase. It is worth noting that once the training is achieved, two pieces of scenario are generated: one contains the assembly of boxes that are necessary to extract BP features in the selected frequency bands and the other contains the trained LDA classifier.

**Step 3: Online Use of the BCI** The last phase is the online use of the BCI. The OpenViBE scenario corresponding to this phase is displayed in Figure 9.

In this scenario, we can observe the classical steps of a BCI which are represented as boxes. The measurement of cerebral activity is represented by the generic network acquisition box. The preprocessing step corresponds to two boxes: (1) the temporal filter box, which filters the data in the 3–45 Hz frequency band (here using a Butterworth filter), and (2) the spatial filter box, which applies a discrete surface Laplacian filter to the data (Wolpaw et al., 2002) in order to build two Laplacian channels over the left and right motor cortices. The feature extraction step is represented by the time based epoching box, which builds an EEG segment representing the last second of data, refreshing each 1/16 s, and by the temporal filter, simple DSP, and signal average boxes, which are used to compute the BP features in the frequency bands identified in the previous phase. Here, the simple DSP box allows us to apply any mathematical formula (such as log-transform or squaring) to the incoming data. These features are then aggregated into a feature vector (feature aggregator box). Note that all these boxes for feature extraction are generated and assembled automatically when running the offline training scenarios, and as such do not need to be assembled by hand. The LDA classifier box is the classification step and uses the LDA that was trained during the previous phase. Finally, the output of this classifier is sent through the VRPN server (analog VRPN server box) to the VR application, which translates it into a command used to interact with the VE and to provide feedback to the subject. The XML stimulation player box is used here to generate the instructions, that is, which movement (left or right) the subject has to imagine. Instructions are used here in order to measure the subject performance. This box sends events to the VR application, using the button VRPN server box that will then provide the corresponding stimuli to the subject. It should be noted that, as with most existing BCI, this BCI is synchronous, which means that the user can interact with the application only during specific time periods that are imposed by the system.

**9.1.4 An Online Experiment with the Handball Application.** In order to illustrate the use of our BCI platform and its suitability to design BCI-based interaction devices for VR, we performed a pilot study with two male subjects (23 and 25 years old). They participated in an online experiment with the handball application in order to assess whether BCI implemented with OpenViBE could be used to interact with a VR application. The two subjects had previously participated in a few motor imagery BCI experiments. It was, however, the first time they used this specific BCI-based VR application. The subjects' brain activity was re-



**Figure 9.** OpenViBE scenario for the handball application. This scenario performs the online processing of the recorded EEG data in order to identify left or right imagined hand movements. The output of this processing is sent to the VR application by using VRPN.

corded using 10 EEG channels (FC3, FC4, C5, C3, C1, C2, C4, C6, CP3, CP4) located over the left and right motor cortices, using a NeXus32B EEG machine from the MindMedia company. The experiment took place in an immersive virtual reality room equipped with a 3 m curved wall on which the VE was projected. The subjects were equipped with stereoscopic glasses. The VE was displayed at a frame rate of 96 Hz.

The two subjects first participated in sessions during which the EEG signals were recorded and stored (step 1 above). These EEG signals were then used to train the LDA classifier (step 2 above). Once a suitable classifier was obtained, the two subjects participated in two game sessions each, as described in Section 9.1.2 (step 3 above). Subject 1 reached a score of 82.5% (33/40) in his two game sessions, whereas subject 2 reached a performance of 70% (28/40) for the first session and 87.5% (35/40) for the second session. By comparison, the score expected by a randomly performing system would be 50% (20/40). These performance scores suggest that the subjects actually had control over the VR application thanks to the BCI. Both subjects also reported that they found the application really entertaining and motivat-

ing, which is in line with results from the literature reporting that VR can increase the motivation during BCI experiments (Friedman et al., 2007). Naturally, these results should be moderated by the small number of subjects involved, but they still suggest that OpenViBE can be used to design a BCI system for interaction with VE. Further evaluations of this application with more subjects is part of ongoing work.

#### 9.2 The Use-the-Force Application

In addition to the handball VR application, we have developed another VR application based on Open-ViBE. This application is known as the use-the-force application, and is an entertaining VR application inspired by the famous "Star Wars" movie. The aim of this application was to explore the design of self-paced BCI (discussed below) and to further validate the Open-ViBE platform with many users and in real-life conditions, outside of laboratories. In the use-the-force application, subjects could lift a virtual spaceship (a TIE fighter) by performing real or imagined foot movements. Indeed, it is well known that, briefly after a real or imagined foot movement, a specific brain signal is generated in the user's brain: an event related synchronization (ERS) in the  $\beta$  rhythm, that is, a brisk increase of EEG amplitude in the 16-24 Hz frequency band (Pfurtscheller, 1999). Interestingly, this brain signal is mainly located in the central area of the brain, and is therefore potentially detectable with a single electrode (electrode Cz).

Using OpenViBE, we have designed a BCI system that can detect this  $\beta$  ERS in electrode Cz, in a selfpaced way, that is, at any time and not only during specific periods. This BCI simply consists of the estimation of a band power feature in the 16–24 Hz band, followed by a comparison of this feature value with a threshold in order to detect whether an ERS occurred. In the VR application, each time an ERS was detected, the virtual spaceship was lifted up at a speed proportional to the amplitude of the ERS. Figure 10 illustrates this application in action in an immersive VR room.

We have evaluated this application with 21 subjects who had no previous BCI experience, during a VR exhibition, that is, in real-life conditions, with a very noisy



**Figure 10.** The use-the-force application. In this application, the user can lift a virtual spaceship by performing real or imagined foot movements. (©CNRS Photothèque/Hubert Raguet)

environment. Our results showed that despite the use of a single electrode and a simple BCI, and despite the fact that the subjects were naive, untrained, and in a very noisy environment, more than half of them were able to control the virtual spaceship using real foot movement from the very first time, and similarly a quarter of them could control it using imagined movement, from the very first time. More details about this experiment can be found in Lotte et al. (2008). In summary, the conducted experiments with this second VR application showed the capability of the OpenViBE platform to design BCI and to use them in real-life conditions. Moreover, regarding the challenging conditions of the experiment (a single EEG channel, no subject training, very noisy environment, etc.), the results obtained appeared to be very promising.

#### **10** Performance Tests

In order to evaluate OpenViBE performances during online operation, we used two different scenarios that were run on three different hardware configurations. Configuration A is an Intel(R) Xeon(TM) CPU 3.80 GHz computer with 4 GB of RAM and was running GNU/Linux Fedora Core 6. Configuration B is an Intel<sup>R</sup> Core<sup>™</sup> 2 CPU T7400 2.16 GHz laptop with 2 GB of RAM and was running GNU/Linux Fedora Core 5. Configuration C is an Intel<sup>R</sup> Core<sup>™</sup> 2 DUO CPU E6850 3 GHz computer with 4 GB of RAM and was running GNU/Linux Ubuntu 9.04 Jaunty.

The first scenario is the handball VR application scenario, which represents a realistic implementation of a BCI. Indeed, the BCI used in the handball VR application consists of frequency and spatial filtering as preprocessing, followed by feature extraction with band-power estimation, and completed by an LDA as the classifier. This design corresponds to well-known BCI systems such as the Graz BCI (Ramoser, Müller-Gerking, & Pfurtscheller, 2000; Pfurtscheller & Neuper, 2001) or the Berlin BCI (Blankertz et al., 2006). The main difference between these two BCI and the handball application BCI lies in the spatial filter used: the Graz-BCI uses bipolar and Laplacian derivations, while our BCI uses a Laplacian derivation and the Berlin BCI uses the Common Spatial Patterns (CSP) algorithm. However, these three spatial filters are all simple linear spatial filters and, as such, require similar computation times. Further differences between these three BCI exist in the machine learning algorithms employed to calibrate these BCI. However, as machine learning for BCI calibration is an off-line operation, it is not of concern here. The OpenViBE scenario for the BCI of the handball application is composed of 34 OpenViBE boxes. It consists of processing 11 channels (10 EEG channels + a reference channel) sampled at 512 Hz and acquired in blocks of 32 samples. The signal processing pipeline is identical to the one described in Section 9.1.1. The average processor load was computed every second over the course of 5 min (300 measures). The global average over the 5 min is presented in Table 1, for each configuration.

In the second scenario, we tried to reach the limits of the platform. The scenario consisted of reading a 512 channel EEG file followed by multiple Butterworth band pass filters. We added as many band pass filters as possible while still keeping the processor load below 100%. Such a scenario could be used when analyzing multiple frequency bands for a large number of chan-

Processor Load Under Different Hardware Configurations			
		Maximum number	
	Processor	of filters	Processor
	load on	on	load on
Computer	Scenario	Scenario	Scenario
configuration	1 (%)	2	2 (%)
А	7.17	6,144	98.26
В	6.80	7,680	88.13

3.45

19,896

97.93

С

**Table 1.** Performance Tests: Processor Load of Scenario 1 and

 Maximum Number of Filters of Scenario 2 with Corresponding

 Processor Load Under Different Hardware Configurations

nels, for example, to design a magnetoencephalographybased BCI (Mellinger et al., 2007). Indeed, MEG systems are generally composed of hundreds of channels. As in the first scenario, the average processor load was computed every second over the course of 5 min. The number of filters we were able to process in real time and the associated global processor load average are displayed in Table 1.

Taken together, our results suggest that our system is able to address realistic use cases such as a motor imagery-based BCI. They also show that our system is able to apply a large number of signal processing algorithms (e.g., band pass filters) while still keeping the real-time constraints.

#### II Current State of the Platform

The OpenViBE software can be downloaded for free at http://openvibe.inria.fr under the terms of L-GPL.<sup>6</sup> The software currently runs on Microsoft Windows 2000/XP/Vista/7 and GNU/Linux. Several acquisition devices are already supported. Those include, for instance, Brainamp Standard, g.Tec g.US-Bamp, MindMedia NeXus32B, MicroMed IntraEEG,

<sup>6.</sup> Plug-ins relying on GPL are available separately under GPL terms.

and CTF/VSF MEG.<sup>7</sup> Existing boxes include generic network acquisition, file reading and writing, signal processing and filtering, feature extraction, and basic classifications, in addition to most common visualization paradigms (e.g., raw signals, spectra, time frequency analysis, 2D/3D topography). OpenViBE also provides the necessary tools to easily use 3D objects in immersive VE, within the platform, in order to design advanced feedback or visualizations. The software is delivered with ready-to-use scenarios. For instance, it proposes a scenario of a BCI based on imagined hand movements that could immediately be used as an interaction device for various applications in real and virtual environments.

#### **12** Conclusion

In this paper, we have presented the OpenViBE platform, a free and open source platform to design, test, and use BCIs in real or virtual environments. Our platform provides the necessary tools for realtime data acquisition, processing, and display of brain signals. The key features of the platform are (1) high modularity, (2) embedded tools for visualization and feedback based on VR and 3D displays, (3) BCI design made available to non-programmers thanks to the OpenViBE designer that enables users to set up a complete BCI without writing a single line of code, and (4) various tools offered to the different types of user, such as the acquisition server or the preconfigured scenarios. OpenViBE also offers the possibility of easy use as an interaction device with any real or virtual environment. The platform capabilities were illustrated on two entertaining VR applications based on a BCI. In the first application, users could move a virtual ball by imagining hand movements, while in the second one, they could lift a virtual spaceship by using real or imagined foot movements. The evaluation of the platform performance demonstrated the suitability of OpenViBE for real-time applications. We believe that OpenViBE could prove a valuable

and useful tool to design innovative BCI-based interaction devices for both VR and real-life applications. This could include applications such as video games and assistive devices. Interested readers can refer to the OpenViBE website to follow the evolution of the platform and download it for free (INRIA, 2010).

#### **12.1 Future Work**

The OpenViBE platform being open source software is aimed at being continuously improved and extended, hopefully by contributors from many different institutions. Currently, our labs are working to propose new functionalities for the authoring tools in order to increase the productivity of end users (both for authors and for operators). Effort will also be dedicated to the growth of the platform by adding, for example, new algorithms for more efficiently detecting a higher number of signals or new visualization techniques of brain activity in VR. Lastly, distributed computing over dedicated hardware will be implemented in the kernel thanks to the box concept.

#### Acknowledgments

This work was supported by the French National Research Agency within the OpenViBE project and grant ANR05RNTL01601. During this work, Fabien Lotte was at INRIA Rennes, France. He now is at the Institute for Infocomm Research (I2R), Singapore. The authors would like to thank Dr. Marc Christie for his fruitful comments on this manuscript.

#### References

- Arrouët, C., Congedo, M., Marvie, J. E., Lamarche, F., Lécuyer, A., & Arnaldi, B. (2005). Open-ViBE: A 3D platform for real-time neuroscience. *Journal of Neurotherapy*, 9(1), 3–25.
- Baillet, S., Mosher, J., & Leahy, R. (2001). Electromagnetic brain mapping. *IEEE Signal Processing Magazine*, 18(6), 14–30.
- Bashashati, A., Fatourechi, M., Ward, R. K., & Birch, G. E.

<sup>7.</sup> A complete list of supported devices can be found on the Open-ViBE website http://openvibe.inria.fr.

(2007). A survey of signal processing algorithms in braincomputer interfaces based on electrical brain signals. *Journal of Neural Engineering*, 4(2), R35–R57.

Blankertz, B., Dornhege, G., Krauledat, M., Müller, K.-R., Kunzmann, V., Losch, F., et al. (2006). The Berlin brain– computer interface: EEG-based communication without subject training. *IEEE Transactions on Neural System Rehabilitation Engineering*, 14(2), 147–152.

Burdea, G. (1996). *Force and touch feedback for virtual reality*. New York: John Wiley & Sons.

Donchin, E., Spencer, K., & Wijesinghe, R. (2000). The mental prosthesis: Assessing the speed of a P300-based brain– computer interface. *IEEE Transactions on Rehabilitation Engineering*, 8(2), 174–179.

Farwell, L., & Donchin, E. (1988). Talking off the top of your head: Toward a mental prosthesis utilizing eventrelated brain potentials. *Electroencephalography and Clinical Neurophysiology*, 70, 510–523.

Friedman, D., Leeb, R., Guger, C., Steed, A., Pfurtscheller, G., & Slater, M. (2007). Navigating virtual reality by thought: What is it like? *Presence: Teleoperators and Virtual Environments*, 16(1), 100–110.

INRIA. (2010). *Open-ViBE platform*. Retrieved January 29, 2010, from http://openvibe.inria.fr/

Krepki, R., Blankertz, B., Curio, G., & Müller, K. R. (2007).
The Berlin brain–computer interface (BBCI): Towards a new communication channel for online control in gaming applications. *Journal of Multimedia Tools and Applications*, 33(1), 73–90.

Lécuyer, A., Lotte, F., Reilly, R., Leeb, R., Hirose, M., & Slater, M. (2008). Brain–computer interfaces, virtual reality and videogames. *IEEE Computer*, 41(10), 66–72.

Leeb, R., Keinrath, C., Friedman, D., Guger, C., Scherer, R., Neuper, C., et al. (2006). Walking by thinking: The brainwaves are crucial, not the muscles! *Presence: Teleoperators and Virtual Environments*, *15*(5), 500–551.

Leeb, R., Scherer, R., Friedman, D., Lee, F., Keinrath, C., Bischof, H., et al. (2007). Combining BCI and virtual reality: Scouting virtual worlds. In G. Dornhege, J. del R. Millan, T. Hinterberger, D. J. McFarland, & K. R. Müller (Eds.), *Towards brain-computer interfacing* (pp. 393–408). Cambridge, MA: MIT Press.

Lotte, F., Congedo, M., Lécuyer, A., Lamarche, F., & Arnaldi, B. (2007). A review of classification algorithms for EEG-based brain-computer interfaces. *Journal of Neural Engineering*, 4, R1–R13.

Lotte, F., Lécuyer, A., Lamarche, F., & Arnaldi, B. (2007).

Studying the use of fuzzy inference systems for motor imagery classification. *IEEE Transactions on Neural System and Rehabilitation Engineering*, 15(2), 322–324.

Lotte, F., Renard, Y., & Lécuyer, A. (2008). Self-paced braincomputer interaction with virtual worlds: A qualitative and quantitative study "out-of-the-lab." In *The 4th International Brain–Computer Interface Workshop and Training Course* (pp. 373–378).

Maggi, L., Parini, S., Perego, P., & Andreoni, G. (2008). BCI++: An object-oriented BCI prototyping framework. In *The 4th International Brain–Computer Interface Work-shop 2008.* 

Mellinger, J., & Schalk, G. (2007). BCI2000: A generalpurpose software platform for BCI. In G. Dornhege, J. R. Millán, et al. (Eds.), *Towards brain-computer interfacing* (pp. 372–381). Cambridge, MA: MIT Press.

Mellinger, J., Schalk, G., Braun, C., Preissl, H., Rosenstiel, W., Birbaumer, N., et al. (2007). An MEG-based brain– computer interface (BCI). *Neuroimage*, 36(3), 581–593.

Millán, J. (2008). Brain-controlled robots. *IEEE Intelligent* Systems, 23(3), 74–76.

Neuper, C., Scherer, R., Wriessnegger, S., & Pfurtscheller, G. (2009). Motor imagery and action observation: Modulation of sensorimotor brain rhythms during mental control of a brain–computer interface. *Clinical Neurophysiology*, 120(2), 239–247.

Nijholt, A. (2009). BCI for games: A "state of the art" survey. In *ICEC 2008*, 225–228.

Pfurtscheller, G. (1999). EEG event-related desynchronization (ERD) and event-related synchronization (ERS). In E. Niedermeyer & F. H. Lopes da Silva, *Electroencephalography: Basic principles, clinical applications and related fields* (4th ed.), pp. 958–967. Baltimore: Williams & Wilkins.

Pfurtscheller, G., & Neuper, C. (2001). Motor imagery and direct brain–computer communication. *Proceedings of the IEEE*, *89*(7), 1123–1134.

Ramoser, H., Müller-Gerking, J., & Pfurtscheller, G. (2000). Optimal spatial filtering of single trial EEG during imagined hand movement. *IEEE Transactions on Rehabilitation Engineering*, 8(4), 441–446.

Sauvan, J., Lécuyer, A., Lotte, F., & Casiez, G. (2009). A performance model of selection techniques for P300-based brain– computer interfaces. In ACM SIGCHI Conference on Human Factors in Computing Systems (ACM CHI), 2205–2208.

Schlögl, A. (2006). GDF—A general dataformat for biosignals. ArXiv [Online journal] Article arXiv:cs/0608052v6.

Schlögl, A., Brunner, C., Scherer, R., & Glatz, A. (2007).

Renard et al. 53

BioSig— Towards brain–computer interfacing. In G. Dornhege, J. R. Millan, T. Hinterberger, D. J. McFarland, & K.-R. Müller (Eds.), *An open source software library for BCI research* (pp. 347–358). Cambridge, MA: MIT Press.

- Taylor, R. M., Hudson, T. C., Seeger, A., Weber, H., Juliano, J., & Helser, A. (2001). VRPN: A device-independent, network-transparent VR peripheral system. In ACM Symposium on Virtual Reality Software and Technology, 55–61.
- Wolpaw, J., Birbaumer, N., McFarland, D., Pfurtscheller, G.,
  & Vaughan, T. (2002). Brain–computer interfaces for communication and control. *Clinical Neurophysiology*, *113*(6), 767–791.
- Zhong, M., Lotte, F., Girolami, M., & Lécuyer, A. (2008). Classifying EEG for brain computer interfaces using Gaussian processes. *Pattern Recognition Letters*, 29, 354– 359.

#### This article has been cited by:

- 1. Rob Zink, Borbála Hunyadi, Sabine Van Huffel, Maarten De Vos. 2016. Tensor-based classification of an auditory mobile BCI without a subject-specific calibration phase. *Journal of Neural Engineering* **13**, 026005. [CrossRef]
- 2. Mengfan Li, Wei Li, Huihui Zhou. 2016. Increasing N200 Potentials Via Visual Stimulus Depicting Humanoid Robot Behavior. International Journal of Neural Systems 26, 1550039. [CrossRef]
- 3. Sangtae Ahn, Kiwoong Kim, Sung Chan Jun. 2016. Steady-State Somatosensory Evoked Potential for Brain-Computer Interface —Present and Future. *Frontiers in Human Neuroscience* 9. . [CrossRef]
- 4. H. Banville, T.H. Falk. 2016. Recent advances and open challenges in hybrid brain-computer interfacing: a technological review of non-invasive human research. *Brain-Computer Interfaces* **3**, 9-46. [CrossRef]
- 5. Jérémy Frey, Aurélien Appriou, Fabien Lotte, Martin Hachet. 2016. Classifying EEG Signals during Stereoscopic Visualization to Estimate Visual Comfort. *Computational Intelligence and Neuroscience* 2016, 1-11. [CrossRef]
- 6. Athanasios Vourvopoulos, Sergi Bermudez i BadiaUsability and Cost-effectiveness in Brain-Computer Interaction 1-8. [CrossRef]
- Regina Gregori Grgič, Enrico Calore, Claudio de'Sperati. 2016. Covert enaction at work: Recording the continuous movements of visuospatial attention to visible or imagined targets by means of Steady-State Visual Evoked Potentials (SSVEPs). Cortex 74, 31-52. [CrossRef]
- 8. Ilkka Kosunen, Mikko Salminen, Simo Järvelä, Antti Ruonala, Niklas Ravaja, Giulio JacucciRelaWorld 208-217. [CrossRef]
- 9. Ignas Martišius, Robertas Damaševičius. 2016. A Prototype SSVEP Based Real Time BCI Gaming System. *Computational Intelligence and Neuroscience* 2016, 1-15. [CrossRef]
- 10. Matthew Dyson, Eoin Thomas, Laurence Casini, Boris Burle. 2015. Online extraction and single trial analysis of regions contributing to erroneous feedback detection. *NeuroImage* 121, 146-158. [CrossRef]
- 11. Dan Zhang, Bisheng Huang, Wei Wu, Siliang Li. 2015. An Idle-State Detection Algorithm for SSVEP-Based Brain–Computer Interfaces Using a Maximum Evoked Response Spatial Filter. *International Journal of Neural Systems* 25, 1550030. [CrossRef]
- 12. Rafael Ramirez, Manel Palencia-Lefler, Sergio Giraldo, Zacharias Vamvakousis. 2015. Musical neurofeedback for treating depression in elderly people. *Frontiers in Neuroscience* 9. . [CrossRef]
- Bastian Venthur, Sven D\u00e4hne, Johannes H\u00f6hne, Hendrik Heller, Benjamin Blankertz. 2015. Wyrm: A Brain-Computer Interface Toolbox in Python. *Neuroinformatics* 13, 471-486. [CrossRef]
- 14. Fabien Lotte, Laurent Bougrain, Maureen ClercElectroencephalography (EEG)-Based Brain-Computer Interfaces 1-20. [CrossRef]
- 15. Catharina Zich, Stefan Debener, Maarten De Vos, Stella Frerichs, Stefanie Maurer, Cornelia Kranczioch. 2015. Lateralization patterns of covert but not overt movements change with age: An EEG neurofeedback study. *NeuroImage* 116, 80-91. [CrossRef]
- 16. Catharina Zich, Stefan Debener, Cornelia Kranczioch, Martin G. Bleichner, Ingmar Gutberlet, Maarten De Vos. 2015. Real-time EEG feedback during simultaneous EEG–fMRI identifies the cortical signature of motor imagery. *NeuroImage* 114, 438-447. [CrossRef]
- 17. Catharina Zich, Maarten De Vos, Cornelia Kranczioch, Stefan Debener. 2015. Wireless EEG with individualized channel layout enables efficient motor imagery training. *Clinical Neurophysiology* **126**, 698-710. [CrossRef]
- 18. Amin Zammouri, Abdelaziz Aitmoussa, Sylvain Chevallier, Eric MonacelliIntelligentocular artifacts removal in a noninvasive singlechannel EEG recording 1-5. [CrossRef]
- 19. B. Colombet, M. Woodman, J.M. Badier, C.G. Bénar. 2015. AnyWave: A cross-platform and modular software for visualizing and processing electrophysiological signals. *Journal of Neuroscience Methods* 242, 118-126. [CrossRef]
- 20. Jérémie Mattout, Margaux Perrin, Olivier Bertrand, Emmanuel Maby. 2015. Improving BCI performance through co-adaptation: Applications to the P300-speller. *Annals of Physical and Rehabilitation Medicine* **58**, 23-28. [CrossRef]
- 21. Wei Li, Mengfan Li, Jing Zhao. 2015. Control of humanoid robot via motion-onset visual evoked potentials. *Frontiers in Systems Neuroscience* 8. . [CrossRef]
- 22. Evan M. Peck, Eleanor Easse, Nick Marshall, William Stratton, L. Felipe PerroneFlyLoop 152-157. [CrossRef]
- 23. Hatma Suryotrisongko, Febriliyan Samopa. 2015. Evaluating OpenBCI Spiderclaw V1 Headwear's Electrodes Placements for Brain-Computer Interface (BCI) Motor Imagery Application. *Procedia Computer Science* 72, 398-405. [CrossRef]
- 24. Sumit Soman, B.K. Murthy. 2015. Using Brain Computer Interface for Synthesized Speech Communication for the Physically Disabled. *Procedia Computer Science* 46, 292-298. [CrossRef]
- 25. Mohit Agarwal, Raghupathy SivakumarTHINK 41-45. [CrossRef]

- 26. Muhammad N. Fakhruzzaman, Edwin Riksakomara, Hatma Suryotrisongko. 2015. EEG Wave Identification in Human Brain with Emotiv EPOC for Motor Imagery. *Procedia Computer Science* 72, 269-276. [CrossRef]
- 27. Vojkan Mihajlovic, Bernard Grundlehner, Ruud Vullers, Julien Penders. 2015. Wearable, Wireless EEG Solutions in Daily Life Applications: What are we Missing?. *IEEE Journal of Biomedical and Health Informatics* 19, 6-21. [CrossRef]
- 28. Raymundo Cassani, Hubert Banville, Tiago H. FalkMuLES 9-12. [CrossRef]
- 29. Daiki Aminaka, Shoji Makino, Tomasz M. RutkowskiChromatic SSVEP BCI paradigm targeting the higher frequency EEG responses 1-7. [CrossRef]
- Roland Ossmann, Stefan Parker, David Thaller, Karol Pecyna, Alvaro García-Soler, Blanca Morales, Christoph Weiß, Christoph Veigl, Konstantinos Kakousis. 2014. As TeRICS, a flexible AT construction set. *International Journal of Adaptive Control and Signal Processing* 28:10.1002/acs.v28.12, 1475-1503. [CrossRef]
- 31. Giljae Lee, Andréa Matsunaga, Salvador Dura-Bernal, Wenjie Zhang, William W Lytton, Joseph T Francis, José A B Fortes. 2014. Towards real-time communication between in vivo neurophysiological data sources and simulator-based brain biomimetic models. *Journal of Computational Surgery* 1. [CrossRef]
- 32. David J. White, Marco Congedo, Joseph Ciorciari. 2014. Source-based neurofeedback methods using EEG recordings: training altered brain activity in a functional brain source derived from blind source separation. *Frontiers in Behavioral Neuroscience* 8. . [CrossRef]
- 33. Parth Gargava, Karan Sindwani, Sumit SomanControlling an arduino robot using Brain Computer Interface 1-5. [CrossRef]
- 34. Berrenur Saylam, Ozan Caglayan, Reis Burak ArslanAnalysis of visual stimulus parameters for SSVEP based brain computer interfaces 1-4. [CrossRef]
- 35. Hubert Cecotti, Bertrand Rivet. 2014. Correction: Cecotti, H. and Rivet, B. Subject Combination and Electrode Selection in Cooperative Brain-Computer Interface Based on Event Related Potentials. Brain Sci. 2014, 4, 335–355. Brain Sciences 4, 488-508. [CrossRef]
- 36. Roberto Meattini, Umberto Scarcia, Claudio Melchiorri, Tony BelpaemeGestural art: A Steady State Visual Evoked Potential (SSVEP) based Brain Computer Interface to express intentions through a robotic hand 211-216. [CrossRef]
- Minkyu Ahn, Mijin Lee, Jinyoung Choi, Sung Jun. 2014. A Review of Brain-Computer Interface Games and an Opinion Survey from Researchers, Developers and Users. Sensors 14, 14601-14633. [CrossRef]
- 38. Emmanuel K. Kalunga, Sylvain Chevallier, Olivier Rabreau, Eric MonacelliHybrid interface: Integrating BCI in multimodal human-machine interfaces 530-535. [CrossRef]
- Christian MÃ<sup>1</sup>/<sub>4</sub>hl, Camille Jeunet, Fabien Lotte. 2014. EEG-based workload estimation across affective contexts. Frontiers in Neuroscience 8. . [CrossRef]
- 40. Damien Petit, Pierre Gergondet, Andrea Cherubini, Maxime Meilland, Andrew I. Comport, Abderrahmane KheddarNavigation assistance for a BCI-controlled humanoid robot 246-251. [CrossRef]
- 41. Hubert Cecotti, Bertrand Rivet. 2014. Subject Combination and Electrode Selection in Cooperative Brain-Computer Interface Based on Event Related Potentials. *Brain Sciences* 4, 335-355. [CrossRef]
- 42. Darlene A. Lobel, Kendall H. Lee. 2014. Brain Machine Interface and Limb Reanimation Technologies: Restoring Function After Spinal Cord Injury Through Development of a Bypass System. *Mayo Clinic Proceedings* **89**, 708-714. [CrossRef]
- 43. Z. Mahmoodin, W. Mansor, Khuan Y Lee, N. B. MohamadProcessing of electroencephalogram signals using OpenVibe 563-567. [CrossRef]
- 44. Gaëtan Sanchez, Jean Daunizeau, Emmanuel Maby, Olivier Bertrand, Aline Bompas, Jérémie Mattout. 2014. Toward a New Application of Real-Time Electrophysiology: Online Optimization of Cognitive Neurosciences Hypothesis Testing. *Brain Sciences* 4, 49-72. [CrossRef]
- 45. Jonathan Mercier-Ganady, Fabien Lotte, Emilie Loup-Escande, Maud Marchal, Anatole LecuyerThe Mind-Mirror: See your brain in action in your head using EEG and augmented reality 33-38. [CrossRef]
- 46. Jane E. Huggins, Christoph Guger, Brendan Allison, Charles W. Anderson, Aaron Batista, Anne-Marie Brouwer, Clemens Brunner, Ricardo Chavarriaga, Melanie Fried-Oken, Aysegul Gunduz, Disha Gupta, Andrea Kübler, Robert Leeb, Fabien Lotte, Lee E. Miller, Gernot Müller-Putz, Tomasz Rutkowski, Michael Tangermann, David Edward Thompson. 2014. Workshops of the Fifth International Brain-Computer Interface Meeting: Defining the Future. *Brain-Computer Interfaces* 1, 27-49. [CrossRef]
- 47. Cornelia Kranczioch, Catharina Zich, Irina Schierholz, Annette Sterr. 2014. Mobile EEG and its potential to promote the theory and application of imagery-based motor rehabilitation. *International Journal of Psychophysiology* **91**, 10-15. [CrossRef]
- 48. Maarten De Vos, Katharina Gandras, Stefan Debener. 2014. Towards a truly mobile auditory brain–computer interface: Exploring the P300 to take away. *International Journal of Psychophysiology* **91**, 46-53. [CrossRef]

- Han-Jeong Hwang, Soyoun Kim, Soobeom Choi, Chang-Hwan Im. 2013. EEG-Based Brain-Computer Interfaces: A Thorough Literature Survey. *International Journal of Human-Computer Interaction* 29, 814-826. [CrossRef]
- 50. Umut Orhan, Deniz Erdogmus, Brian Roark, Barry Oken, Melanie Fried-Oken. 2013. Offline analysis of context contribution to ERP-based typing BCI performance. *Journal of Neural Engineering* **10**, 066003. [CrossRef]
- 51. Jing Zhao, Qinghao Meng, Wei Li, Mengfan Li, Fuchun Sun, Genshe ChenAn OpenViBE-based brainwave control system for Cerebot 1169-1174. [CrossRef]
- 52. Romain Trachel, Thomas Brochier, Maureen ClercAdaptive and warning displays with Brain-Computer Interfaces: Enhanced visuospatial attention performance 367-370. [CrossRef]
- 53. Griffin Milsap, Matthew Fifer, Nathan Crone, Nitish ThakorListening to the music of the brain: Live analysis of ECoG recordings using digital audio workstation software 682-685. [CrossRef]
- 54. Alexandra-Maria Tautan, Wouter Serdijn, Vojkan Mihajlovic, Bernard Grundlehner, Julien PendersFramework for evaluating EEG signal quality of dry electrode recordings 186-189. [CrossRef]
- 55. Athanasios Vourvopoulos, Ana Lucia Faria, Monica S Cameirao, Sergi Bermudez i BadiaRehabNet: A distributed architecture for motor and cognitive neuro-rehabilitation 454-459. [CrossRef]
- 56. Christian Andreas Kothe, Scott Makeig. 2013. BCILAB: a platform for brain-computer interface development. *Journal of Neural Engineering* 10, 056014. [CrossRef]
- 57. L. Mayaud, M. Congedo, A. Van Laghenhove, D. Orlikowski, M. Figère, E. Azabou, F. Cheliout-Heraut. 2013. A comparison of recording modalities of P300 event-related potentials (ERP) for brain-computer interface (BCI) paradigm. *Neurophysiologie Clinique/Clinical Neurophysiology* 43, 217-227. [CrossRef]
- 58. Tomas Skripcak, Pavol Tanuska, Uwe Konrad, Nils Schmeisser. 2013. Toward Nonconventional Human–Machine Interfaces for Supervisory Plant Process Monitoring. *IEEE Transactions on Human-Machine Systems* 43, 437-450. [CrossRef]
- Christof Pieloth, Jorge M. Pizarro, Thomas Knosche, Burkhard Maess, Mirco FuchsAn online system for neuroelectromagnetic source imaging 270-274. [CrossRef]
- 60. Emmanuel Kalunga, Karim Djouani, Yskandar Hamam, Sylvain Chevallier, Eric MonacelliSSVEP enhancement based on Canonical Correlation Analysis to improve BCI performances 1-5. [CrossRef]
- 61. Subhadeep KaranA literature survey on the contemporary methodologies used in brain computer interface for spelling application 1-5. [CrossRef]
- 62. J. Legeny, R. Viciana-Abad, A. Lecuyer. 2013. Toward Contextual SSVEP-Based BCI Controller: Smart Activation of Stimuli and Control Weighting. *IEEE Transactions on Computational Intelligence and AI in Games* 5, 111-116. [CrossRef]
- 63. Eoin Thomas, Joan Fruitet, Maureen Clerc. 2013. Combining ERD and ERS features to create a system-paced BCI. *Journal of Neuroscience Methods* 216, 96-103. [CrossRef]
- 64. Laurent Bonnet, Fabien Lotte, Anatole Lecuyer. 2013. Two Brains, One Game: Design and Evaluation of a Multiuser BCI Video Game Based on Motor Imagery. *IEEE Transactions on Computational Intelligence and AI in Games* 5, 185-198. [CrossRef]
- 65. Chantri Polprasert, Pratana Kukieattikool, Tanee Demeechai, James A Ritcey, Siwaruk Siwamogsatham. 2013. New stimulation pattern design to improve P300-based matrix speller performance at high flash rate. *Journal of Neural Engineering* **10**, 036012. [CrossRef]
- 66. N. Kos'myna, F. Tarpin-Bernard. 2013. Evaluation and Comparison of a Multimodal Combination of BCI Paradigms and Eye Tracking With Affordable Consumer-Grade Hardware in a Gaming Context. *IEEE Transactions on Computational Intelligence and AI in Games* 5, 150-154. [CrossRef]
- 67. Ueli Rutishauser, Andreas Kotowicz, Gilles Laurent. 2013. A method for closed-loop presentation of sensory stimuli conditional on the internal brain-state of awake animals. *Journal of Neuroscience Methods* **215**, 139-155. [CrossRef]
- 68. L. Mayaud, S. Filipe, L. Pétégnief, O. Rochecouste, M. Congedo. 2013. Robust Brain-computer Interface for virtual Keyboard (RoBIK): Project results. *IRBM* 34, 131-138. [CrossRef]
- 69. Joan Fruitet, Alexandra Carpentier, Rémi Munos, Maureen Clerc. 2013. Automatic motor task selection via a bandit algorithm for a brain-controlled button. *Journal of Neural Engineering* **10**, 016012. [CrossRef]
- 70. C. Veigl, C. Weis, K. Kakousis, D. Ibanez, A. Soria-Frisch, A. CarboneModel-based design of novel human-computer interfaces The Assistive Technology Rapid Integration & Construction Set (AsTeRICS) 1-7. [CrossRef]
- 71. Matthieu Duvinage, Thierry Castermans, Mathieu Petieau, Thomas Hoellinger, Guy Cheron, Thierry Dutoit. 2013. Performance of the Emotiv Epoc headset for P300-based applications. *BioMedical Engineering OnLine* **12**, 56. [CrossRef]

- 72. Stefan Debener, Falk Minow, Reiner Emkes, Katharina Gandras, Maarten de Vos. 2012. How about taking a low-cost, small, and wireless EEG for a walk?. *Psychophysiology* **49**:10.1111/psyp.2012.49.issue-11, 1617-1621. [CrossRef]
- 73. Pierre Gergondet, Damien Petit, Abderrahmane KheddarSteering a robot with a brain-computer interface: Impact of video feedback on BCI performance 271-276. [CrossRef]
- 74. E. Thomas, J. Fruitet, M. ClercInvestigating brief motor imagery for an ERD/ERS based BCI 2929-2932. [CrossRef]
- 75. B. Venthur, B. BlankertzMushu, a free- and open source BCI signal acquisition, written in Python 1786-1788. [CrossRef]
- 76. G. Charvet, M. Foerster, G. Chatalic, A. Michea, J. Porcherot, S. Bonnet, S. Filipe, P. Audebert, S. Robinet, V. Josselin, J. Reverdy, R. D'Errico, F. Sauter, C. Mestais, A. L. BenabidA wireless 64-channel ECoG recording Electronic for implantable monitoring and BCI applications: WIMAGINE 783-786. [CrossRef]
- 77. Soumya Sen Gupta, Sumit Soman, P. Govind Raj, Rishi Prakash, S. Sailaja, Rupam BorgohainDetecting eye movements in EEG for controlling devices 69-73. [CrossRef]
- 78. Matthew Dyson, Laurence Casini, Boris BurleDiscrimination of Discrete Feedback During Performance of Motor Imagery 93-96. [CrossRef]
- 79. Matthieu Duvinage, Thierry Castermans, Mathieu Petieau, Guy Cheron, Thierry DutoitAre current gait-related artifact removal techniques useful for low-complexity BCIs? 1-7. [CrossRef]
- 80. C. Breitwieser, I. Daly, C. Neuper, Gernot R. Muller-Putz. 2012. Proposing a Standardized Protocol for Raw Biosignal Transmission. *IEEE Transactions on Biomedical Engineering* 59, 852-859. [CrossRef]
- Nicolas Brodu, Fabien Lotte, Anatole Lécuyer. 2012. Exploring two novel features for EEG-based brain-computer interfaces: Multifractal cumulants and predictive complexity. *Neurocomputing* 79, 87-94. [CrossRef]
- 82. Daniela Ferreira, Nuno DiasIdentification of performance-related EEG phenotypes for neurofeedback training 1-7. [CrossRef]
- 83. G. Cheron, M. Duvinage, C. De Saedeleer, T. Castermans, A. Bengoetxea, M. Petieau, K. Seetharaman, T. Hoellinger, B. Dan, T. Dutoit, F. Sylos Labini, F. Lacquaniti, Y. Ivanenko. 2012. From Spinal Central Pattern Generators to Cortical Network: Integrated BCI for Walking Rehabilitation. *Neural Plasticity* 2012, 1-13. [CrossRef]
- 84. Emmanuel Maby, Margaux Perrin, Olivier Bertrand, Gaëtan Sanchez, Jérémie Mattout. 2012. BCI Could Make Old Two-Player Games Even More Fun: A Proof of Concept with "Connect Four". Advances in Human-Computer Interaction 2012, 1-8. [CrossRef]
- 85. Perrin Margaux, Maby Emmanuel, Daligault Sébastien, Bertrand Olivier, Mattout Jérémie. 2012. Objective and Subjective Evaluation of Online Error Correction during P300-Based Spelling. *Advances in Human-Computer Interaction* 2012, 1-13. [CrossRef]
- 86. Pierre Gergondet, Sebastien Druon, Abderrahmane Kheddar, Christoph Hintermuller, Christoph Guger, Mel SlaterUsing braincomputer interface to steer a humanoid robot 192-197. [CrossRef]
- 87. Thierry Castermans, Matthieu Duvinage, Mathieu Petieau, Thomas Hoellinger, Caty De Saedeleer, Karthik Seetharaman, Ana Bengoetxea, Guy Cheron, Thierry Dutoit. 2011. Optimizing the Performances of a P300-Based Brain–Computer Interface in Ambulatory Conditions. *IEEE Journal on Emerging and Selected Topics in Circuits and Systems* 1, 566-577. [CrossRef]
- 88. C. Breitwieser, C. Neuper, G. R. Muller-PutzA concept to standardize raw biosignal transmission for brain-computer interfaces 6377-6380. [CrossRef]
- 89. L. George, F. Lotte, R. V. Abad, A. LecuyerUsing scalp electrical biosignals to control an object by concentration and relaxation tasks: Design and evaluation 6299-6302. [CrossRef]
- 90. Nicolas Brodu, Fabien Lotte, Anatole LecuyerComparative study of band-power extraction techniques for Motor Imagery classification 1-6. [CrossRef]
- 91. P Brunner, L Bianchi, C Guger, F Cincotti, G Schalk. 2011. Current trends in hardware and software for brain–computer interfaces (BCIs). *Journal of Neural Engineering* **8**, 025001. [CrossRef]
- 92. K. Jerbi, J.R. Vidal, J. Mattout, E. Maby, F. Lecaignard, T. Ossandon, C.M. Hamamé, S.S. Dalal, R. Bouet, J.-P. Lachaux, R.M. Leahy, S. Baillet, L. Garnero, C. Delpuech, O. Bertrand. 2011. Inferring hand movement kinematics from MEG, EEG and intracranial EEG: From brain-machine interfaces to motor rehabilitation. *IRBM* 32, 8-18. [CrossRef]
- 93. Marco Congedo, Leslie SherlinEEG Source Analysis 25-433. [CrossRef]
- 94. Arnaud Delorme, Tim Mullen, Christian Kothe, Zeynep Akalin Acar, Nima Bigdely-Shamlo, Andrey Vankov, Scott Makeig. 2011. EEGLAB, SIFT, NFT, BCILAB, and ERICA: New Tools for Advanced EEG Processing. *Computational Intelligence and Neuroscience* 2011, 1-12. [CrossRef]

- 95. Alan D. Degenhart, John W. Kelly, Robin C. Ashmore, Jennifer L. Collinger, Elizabeth C. Tyler-Kabara, Douglas J. Weber, Wei Wang. 2011. Craniux: A LabVIEW-Based Modular Software Framework for Brain-Machine Interface Research. *Computational Intelligence and Neuroscience* 2011, 1-13. [CrossRef]
- 96. Hubert Cecotti. 2011. Spelling with non-invasive Brain–Computer Interfaces Current and future trends. *Journal of Physiology-Paris* 105, 106-114. [CrossRef]
- 97. Brahim Hamadicharef, Mufeng Xu, Sheel AdityaBrain-Computer Interface (BCI) Based Musical Composition 282-286. [CrossRef]